Dear Reviewers,
First of all, thank you very much for your valuable comments on our manuscript, we have revised and uploaded the manuscript according to the comments. I'm very sorry, because we don't know how to upload pictures on the website, we have to upload the pictures and tables in the form of files, and upload the revised full text to the website together. Secondly, in response to your valuable revision comments, we have made some improvements: 1. We reviewed the latest literature and updated the summary of clinical manifestations, diagnosis and treatment of primary renal small cell carcinoma. 2. We incompletely counted 92 globally published cases of renal SCC from 1984 to 2022 and summarised their clinical characteristics and treatment options. 3. We reviewed the latest systematic reviews and meta-analyses for the treatment of primary renal small cell carcinoma we provide It presents the latest evidence and summarizes current treatment options and their impact on prognosis. 4. For language problems, we handed over the article to professionals to review and make revisions, and the language editing certificate has been uploaded to the website. Finally, thank you again for your contribution to our manuscript and hope our research will be published in your journal!
Best regards,
mnwkdlxk@163.com